Author’s response to reviews

Title: Body size perceptions and preferences favor overweight in adult Saharawi refugees

Authors:
Desire Naigaga (Desire.Naigaga@hioa.no)
David Jahanlu (djahanlu@me.com)
Hanne Marit Claudius (hmclaudius@gmail.com)
Anne Karine Gjerlaug (annekarine.gjerlaug@hioa.no)
Ingrid Barikmo (ingrid.barikmo@outlook.com)
Sigrun Henjum (sigrun.henjum@hioa.no)

Version: 1 Date: 18 Sep 2017

Author’s response to reviews:

Dear Sir!

We have added information on informed consent and Ethical Committee to the manuscript:

"The study was approved by the Saharawi Health Authorities and The Norwegian Regional Committees for Medical and Health Research Ethics (2014/1155). Informed written consent was obtained from all participants and the study was conducted according to the guidelines provided in the Declaration of Helsinki".